Author's response to reviews

Title: A perforated duodenal ulcer presenting with a subphrenic abscess revealed by plain abdominal films and confirmed by MDCT: A case report.

Authors:

LUIGI CAMERA (camera@unina.it)
MILENA CALABRESE (mcalabrese@gmail.com)
VALERIA ROMEO (valeria.romeo@libero.it)
FABRIZIO SCORDINO (fascordi@gmail.com)
PIER PAOLO MAINENTI (pierpamainenti@hotmail.it)
MARCO CLEMENTE (marco.clementa@hotmail.it)
GAETANO RAPICANO (gaetano.rapicano@alice.it)
MARCO SALVATORE (marsalva@unina.it)

Version: 5 Date: 16 August 2013

Author's response to reviews:

REV # 1: The requested minor changes in the text have been made and are underlined and reported in red. The figures are all composite as described in the text (Fig. 1a, b; etc)

REV # 2: The manuscript has been extensively edited and considerably shortened, as requested. The post-operative clinical course has been described in more details and the possible role of Imatinib in determining the peptic ulcer has been elucidated and a reference (#14) was added.